Author’s response to reviews

Title: Performance management methods and practices among Nurses in Primary Health Care Settings: A systematic scoping review protocol

Authors:
Cynthia Madlabana (madlabana@ukzn.ac.za;zmadlabane@gmail.com)
Tivani Mashamba-Thompson (mashamba-thompson@ukzn.ac.za)
Inge Petersen (Peterseni@ukzn.ac.za)

Version: 3 Date: 03 Jul 2019

Author’s response to reviews:

Thank you for the opportunity for us to revise our paper.

we have reworked Line numbers 54-74

Reworked the Contribution to the field section

We have included a definition of high-quality health care as per the Lancet Commission on High Quality Health Systems in the SDG Era – see lines 65-70

The research questions for the scoping review are appropriately place in the methods section. We have removed them from the background where we have rather highlighted the overall aim of the scoping review.